Author's response to reviews

Title: The opportunities for and obstacles against prevention: the example of Germany in the areas of tobacco and alcohol

Authors:

Ulla Walter (Walter.Ulla@MH-Hannover.de)
Marc Suhrcke (m.suhrcke@uea.ac.uk)
Miriam G Gerlich (Gerlich.Miriam@MH-Hannover.de)
Till A Boluarte (till.boluarte@uni-wh.de)

Version: 7 Date: 18 August 2010

Author's response to reviews: see over
Resubmission for BMC Public Health after reformatting
Category “Debate”.

Ms. No.: 6236362423490892

To the Editors
We would like to thank the editorial office for giving us the chance to finally review the article „The opportunities for and obstacles against prevention: the example of Germany in the areas of tobacco and alcohol“.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

We changed one typographic error in the legend of figure 2, it reads now “statutory”. 